Title: Serum Biomarkers for Neurofibromatosis Type 1 and early detection of malignant peripheral nerve sheath tumors

Authors:

Su-Jin Park (su-jin.park@charite.de)
Birgit Sawitzki (birgit.sawitzki@charite.de)
Lan Kluwe (kluwe@uke.de)
Victor F Mautner (v.mautner@uke.uni-hamburg.de)
Nikola Holtkamp (nikola.holtkamp@charite.de)
Andreas Kurtz (andreas.kurtz@charite.de)

Version: 5 Date: 2 March 2013

Author’s response to reviews:

Dear Editor,

We have now revised the manuscript according to the requested editorial changes:

- We have completed the author’s address list and included the Email addresses.
- We have revised the author’s contribution according to the instructions
- We have revised the acknowledgement according to the instructions